



Article Green Port Industry to Support the Offshore Wind Sector: A Proposal Framework

Monalisa Godeiro ¹, Mario González ^{1,*}, Dylan Jones ², Negar Akbari ², Gabriela Nascimento ¹, David Melo ¹, Rafael Vasconcelos ¹, Andressa Santiso ¹, Luana Nogueira ¹, Mariana Almeida ¹ and José Toledo ³

- ¹ CREATION Research Group in Renewable Energies and Power-to-X, Graduate Program in Production Engineering (PEP), Federal University of Rio Grande do Norte (UFRN), Lagoa Nova University Campus, Natal 59078-900, Brazil; monalisa.godeiro@ufrn.br (M.G.); gabriela.soares.060@ufrn.edu.br (G.N.); david.cassimiro.096@ufrn.edu.br (D.M.); rafael.monteiro.050@ufrn.br (R.V.); andressa.santiso.014@ufrn.edu.br (A.S.); luana.nogueira.104@ufrn.edu.br (L.N.); almeidamariana@yahoo.com (M.A.)
- ² Centre for Operational Research and Logistics, University of Portsmouth, Portland Building, Portland Street, Portsmouth PO1 3AH, UK; dylan.jones@port.ac.uk (D.J.); negar.akbari@port.ac.uk (N.A.)
- ³ Quality Study and Research Group (GEPEQ), Graduate Program in Production Engineering, Federal University of São Carlos (UFSCAR), University City, São Carlos 13565-905, Brazil; toledo@dep.ufscar.br
- * Correspondence: mario.gonzalez@ufrn.br

Abstract: In recent years, offshore wind power has become increasingly relevant as a key alternative for contributing to the global economy's decarbonization. Also, the accelerated technological development of the offshore wind turbine influences the increase in size and weight of its main components. This requires an appropriate port infrastructure to support the installation, operation, and maintenance and future decommissioning of offshore wind farms, and especially to serve as an area for manufacturing these components, addressing logistical challenges associated with land transport. This research aims to identify the factors that characterize a suitable port to support the offshore wind industry, also bringing the new green port industry concept. A systematic literature review was conducted via analyses of 126 documents, and a survey procedure was applied to validate the proposed model. As a result, a characterization model was proposed that includes 71 factors classified into 6 dimensions: physical characteristics, port layout, connectivity, port operation, port–farm performance optimization, and governance for sustainability, which is the main novelty of this study. The results contribute to the advancement of the offshore wind energy sector and can provide significant benefits for regional development and local communities with offshore wind potential.

Keywords: offshore wind; offshore wind support ports; port logistics; port infrastructure; green port; systematic literature review

1. Introduction

The generation of energy from renewable sources is essential to neutralize carbon emissions into the environment and for countries to have a clean and renewable energy matrix [1]. That includes offshore wind energy, which is considered an important source in this global energy transition. It is necessary to deploy 2000 GW of offshore wind by 2050 to reach net zero. This requires a huge upsurge in installations, with 35 GW of offshore wind to be added annually in the coming decade, starting from a global total of just over 60 GW [2,3].

By the end of 2022, 64.3 GW of offshore wind capacity had been installed worldwide, with China leading the installed capacity, followed by the UK and other European countries [2]. In addition, new markets have emerged, for instance, in the USA, Japan, Taiwan, Vietnam, India, South Africa, and Brazil [4,5], as a result of cost reductions in



Citation: Godeiro, M.; González, M.; Jones, D.; Akbari, N.; Nascimento, G.; Melo, D.; Vasconcelos, R.; Santiso, A.; Nogueira, L.; Almeida, M.; et al. Green Port Industry to Support the Offshore Wind Sector: A Proposal Framework. *Energies* **2024**, *17*, 6155. https://doi.org/10.3390/en17236155

Academic Editor: Gianmaria Giannini

Received: 19 October 2024 Revised: 29 November 2024 Accepted: 4 December 2024 Published: 6 December 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the rapid technological development of turbines as well as due to improved logistics and supply efficiency throughout the offshore wind farm's lifecycle.

There is a tendency to continue the technological development of the offshore wind turbine and the consequent increase in the dimensions of its components to generate a greater energy quantum, better performance, and higher nominal power, as the increase in height and swept area consequently increases the capacity factor [6,7]. As examples of this evolution, in 2010 there were turbine models with 3 MW and in 2015 with 6 MW of nominal power. In 2018, a 12 MW model was launched by General Electric (GE), with a blade length of 107 m [8]. In 2020, Siemens Gamesa launched a 14 MW turbine, with sales projection for 2024, measuring 108 m in blade length [9]. In 2023, China Three Gorges Corporation had installed the world's first 16 MW offshore wind turbine, off the coast of Fujian, China, with a blade length of 123 m [10].

Consequently, this increase in dimensions and weight brought challenges for transport, handling, and movement of components, which creates the necessity for manufacturers of these large components, e.g., blades, to install their industries in a port area. This would make the role of port infrastructure essential in the development of offshore wind farms, as the main onshore support base throughout all lifecycle phases [11–13].

It was observed that turbines with a nominal capacity above 5 MW have severe restrictions on the transport of components by road or rail [12,13]. These bottlenecks can be mitigated by using ports with layout, connectivity, and physical characteristics [11,14] that add value to the entire chain, optimizing them, and creating an important link between land and maritime activities.

Thus, it is necessary to build or adapt the port infrastructure linked to the concept of the port industry as close as possible to the offshore wind site, so that it is possible to support not only the manufacturing, but also the installation, operation, and maintenance (O&M) of components, such as the blades, nacelle, tower, and foundation. In regions that have the potential to install more than 5 GW of OWF, the construction of a manufacturing port becomes viable and attractive to developers [15].

In addition to the technical factors, there must also be a concern for the sustainability elements to be linked to this port industry. Through the concept of green ports [16], one must be concerned not only with implementing operations to support the clean energy sector, but also with adding governance elements for sustainability, with actions and guidelines that promote this sustainability in these operations, contributing to the reduction of greenhouse gas emissions [16,17].

When analyzing existing studies around ports to support the offshore wind sector, it is noted that there are studies that identify the essential characteristics of a base port, whether it be for the installation or operation and maintenance phases. There is a study that described the port requirements for the offshore wind industry in the context of cooperation between ports in the North Sea, Germany, addressing the characteristics of base ports, and quick reaction/O&M and supply ports [18].

There is another study that addressed key port requirements to support offshore wind development in North America with a brief comparison of traditional ports versus offshore wind ports [19]. Another article proposed a project for a logistic port for offshore wind turbines in the United States and presented five critical elements that must be applied [20].

About the suitable ports, with the aim to rank the most suitable ports for installation and O&M, the authors of [11] evaluated and selected the most suitable port to support an offshore wind farm in the United Kingdom, using the analytic hierarchy process (AHP) method, in which 3 criteria and 15 sub-criteria were classified and prioritized.

Another study defined the optimized layout of an installation port to support an offshore wind farm, in which the port area was segmented into sub-areas: unloading, storage, preparation or assembly areas, and loading areas, in order to minimize the logistics transportation costs within the port [21].

The authors of [14] discussed the need for the supply chain to be ready for a green transformation regarding the challenges of offshore wind logistics and the role of ports.

In 2020, a review study was prepared by the authors of [22] about the installation of offshore wind turbines and, in this context, revealed some elements required of ports to meet this activity.

Therefore, although some countries are advancing in the development of studies, gaps were identified in the academic literature regarding the following: (i) research that systematically addresses the factors that characterize the port infrastructure necessary for the offshore wind industry, (ii) research that details and defines the concept of port industry for the offshore wind sector, and (iii) studies that address the ports in terms of not only environmental sustainability, linked to the concept of green ports, but also socioeconomics, performance optimization, and at the operational level, focused on serving this sector.

Thus, from these gaps, the research question arose: which factors should be considered to characterize an offshore wind support port? This article aims to identify the factors that characterize a suitable port to support the offshore wind sector.

The article is organized into six sections. Section 2 deals with the materials and methods. Section 3 addresses the literature review, with relevant concepts on the subject. Section 4 provides the results about the conceptual model and the proposal framework. Section 5 presents the main discussions. Finally, Section 6 considers the conclusions and recommendations.

2. Materials and Methods

The research is characterized as theoretical [23], with a qualitative–quantitative approach [24], as we used predominantly qualitative data but also quantitative data to analyze the results. We also used inductive logical argumentation [25], as new knowledge was generated from various factors, and a technical procedure for bibliographic research [26,27], as we used the systematic literature review (SLR) to obtain the conceptual framework as well as a survey procedure, applying a questionnaire to validate the framework.

The SLR consists of a systematic, rigorous, and holistic analysis of a given topic, based on scientific evidence, to reduce the bias associated with studies using non-systematic reviews [28–30]. Compared to the traditional review, the SLR declares the purpose of the review and performs a thorough search with combinations of keywords, in addition to being concerned with the possibility of replication [31,32]. Thus, it aims to generate structured knowledge to make predictions about a particular topic [33].

The research procedure included 6 steps, as shown in Figure 1.

STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
Exploratory Theoretical Research	Systematic Literature Review (SLR)	Analysis of technical reports	Development of the conceptual framework	Framework validation and case studies	Analysis of results and conclusions

Figure 1. Steps of the research procedure.

First, we conducted exploratory theoretical research to contextualize the main topics about the offshore wind sector and related port infrastructure. In the second step, we started the systematic literature review (SLR), which was performed in four stages, as shown in Figure 2, including SLR planning, screening and selection of articles, detailed analysis of documents, and framework modeling.

In the first stage, after defining the research question, the combinations of keywords and databases used were defined. The 8 keyword combinations are described in Table 1, which were explored in 8 databases: Capes (www.periodicos.capes.gov.br, accessed on 3 February 2020), Scopus, Science Direct, Web of Science, Emerald, Semantic Scholar, NTNU, and CORE journals, which are known in the research theme. The total number of documents found was 1965.

		SLR	Planning							
Def ke	inition of eywords	De	finition of atabases	Init resu	ial Ilts					
8 con	nbinations	8 da	tabases	n = 196	55					
				Screen	ing and Selection	ı of a	rticles			
	Duplic exclus	cate sion	Exclu being	ision for not from journal	Exclusion for b out of scop	eing e	Insertion by refe- rences and author	s	Total of articles included	
	n = 639	9	n =	= 219	n = 1043		n = 30		n = 94	
						De	tailed analysis of	docu	uments	
					Data collection descriptive an	on for alysis	Data collection the study perfo	fron rmec	n Factors c	ollection
					Authors; Year Country; Stud Approach	;; y;	Summary; Objec Type of search; f studies	ctive; uture	; Characterist e each author dimen	ics cited by in different sions
								Fr	amework mode	lling

	or in mouring	
Descriptive analysis	Content analysis	Conceptual Framework
Publications by year, country, author, by type of approach and study	Analysis of the types of ports; port operation; phases	71 factors classified into 6 dimensions

Figure 2. Stages of the SLR.

Table 1. Keywords and combinations included in the selection of the articles.

Combination	Title Search	Boolean Operator	Search on "All Options"	Results Included
1	Port		Offshore wind	27
2	Ports		Offshore wind	3
3	Requirement		Offshore wind port	1
4	Installation	,	Offshore wind port	15
5	Operation and maintenance	and	Offshore wind port	3
6	Decommissioning		Offshore wind port	4
7	Industry		Offshore wind port	4
8	Green port concept		-	7
	1 1		Reference articles	24
			Other publications by authors	6
			Total	94

Filters were not used to restrict the publication period, so it was between 2001 and 2021. For step 3 of this research procedure, 9 theses, 6 book chapters, and 17 technical reports were included in the SLR to enrich the research through the many factors also mentioned in the reports, totaling 126 documents. All data were catalogued and systematized using the affinity diagram.

Subsequently, in step 4, it was possible to elaborate the framework modeling on the dimensions and factors that a port must have to support the offshore wind sector. The modeling went through the elaboration, critical analysis, and improvement phases, with 5 repetition cycles until reaching the final version. The dimensions were obtained through the reading of technical reports.

After that, it was necessary to validate the developed framework (step 5) through the application of a questionnaire answered by experts in the field of offshore wind ports. The following were considered as specialists: authors with the largest number of publications, authors of the analyzed technical reports, speakers at international events on ports, port managers of the main ports that support the offshore wind sector, and experts in the study of the offshore wind sector. The population considered was 70 specialists, to whom the questionnaire link was sent in 5 rounds. The total period for this phase was 11 days. After

this, the collection, description, and analysis of the responses obtained by the 15 respondents occurred, which corresponds to 21% of the population, and all these respondents came from 7 different countries: Germany, the United Kingdom, Denmark, the Netherlands, Belgium, the United States, and Brazil.

The tool used was Google Forms and the questionnaire was divided into eight sections: an introductory section, six sections, one for each dimension studied and their respected factors identified, and a final section on respondent identification data. In each section on dimensions and factors, the following were asked: a closed question with a scale ranging from 3 answers ("I agree", "I agree in parts", and "I disagree") to facilitate judgment, and two open questions about the justification of disagreement and suggestion of new factors.

Also in step 5, to validate and strengthen the proposal framework, case studies were conducted about three main ports to support the offshore sector in their countries: Port of Esbjerg, Green Port Hull, and Bremerhaven Port. With this inter-case analysis, it was possible to compare the factors and their requirements presented in the framework and the literature and discuss good practices in the operations of each port.

Finally, in step 6, it was possible to analyze the results, considering the knowledge obtained and validations carried out by experts, and propose a model for characterization of a green port industry to serve the offshore wind sector.

3. Literature Review

3.1. Logistics and Importance of Ports

The lifecycle of an offshore wind power plant comprises five main phases: (i) project, with information gathering, development, and concession; (ii) acquisition and manufacturing, with choice of suppliers to acquire and manufacture components; (iii) installation and commissioning, in which all plant components are installed, commissioned, and tested; (iv) operation and maintenance, which corresponds to the operationalization of the park and preventive and corrective maintenance; (v) decommissioning of part or the whole, such as turbines, foundations, cables, and substations [14,34,35].

Throughout this lifecycle, logistics is a fundamental factor, which facilitates the flow of resources needed to carry out the various activities during all phases of an offshore wind project [36,37]. It is, therefore, the storage and movement, on land and sea, of components, people, and equipment, safely and under variable weather conditions, inserted in a supply network with different chains, but seeking maximum efficiency [13]. Thus, three resources are indispensable in the logistical process: ports, vessels, and installation equipment [36].

The role of port logistics becomes increasingly important due two main factors: (i) The tendency to locate more manufacturing facilities and assembly operations in ports, near waterways, due to the size and weight of the turbine's components and, therefore, the logistical challenges associated with its inland transport [13], as they represent the interface between onshore and offshore operations [38–40]. (ii) Ports are also a critical part to bring efficiency during all offshore wind farm lifecycle phases and serve as a land base primarily to support installation, as well as the O&M and decommissioning phases [11,41].

Given not only the high cost of offshore operations but also the risk and uncertainty related to environmental conditions, it is important to carry out as much activity onshore as possible to reduce the number of offshore lifts, cost, and time [42,43]. A scenario projected by WindEurope predicted, by 2030, a total of 70 GW of installed capacity from offshore wind farms in Europe, where ports will have to service 10,000 wind turbines for O&M, install about 460 turbines/year, and decommission 600 turbines/year. Based on this, it is estimated that ports can contribute 5.3% of the total LCOE reduction, which is equivalent to a CAPEX reduction of EUR 185,000/MW. To invest in the construction or adaptation of ports, only 10–20% savings in CAPEX are needed, demonstrating feasibility in its use [44].

3.2. Ports for Offshore Wind: Types and Definitions

A port is a link in the transport chain that promotes economic development in a region through the flow of people and goods between the sea and land [45]. However, over time,

ports have developed other features and, consequently, several types are found in the literature. Figure 3 presents some of the types of ports with their respective definitions, capable of supporting the phases of the lifecycle of offshore wind farms.

Port cluster	A cluster can be defined as a geographic region based on a relevant port location, incorporating most, if not all, of the necessary offshore wind energy value.
Port super hub	It is at the highest capacity level, whose function is to build, assembly, test and transport turbines and foundations, also as an interface for manufacturers.
Manufacturing port	Used during the farm construction phase. Its areas are needed to manufacture the main components and structures, so the value-adding activities can take place in the port.
Installation port	The main components are manufactured and then delivered to the installation port, to be then stored and pre-assembled, for subsequent transport to the offshore installation site.
Base port	Base ports has a wider managerial and organizational use. Also, some authors call those used for the production, storage of components and assembly.
Preparation port	Like installation ports but focussed on preparation or pre-assembly. Components can be shipped from the manufacturing port and prepared before transporting to OWF.
Transportable port	It can be used in several projects within a short distance of them. It is essential to have static stability and verification of movement response.
Assembly port	It has functions like to the preparation ports, focused on aiding the assembly, where large areas are usually designated to minimize offshore operations and lifts as much as possible.
Service port	The rapid reaction port differs from the previous one in that they are intended to support spontaneous and short-term maintenance operations, with quick access. It must guarantee the availability of spare parts and transfer 24 hours a day, 7 days a week.
O&M port	As Operation and Maintenance (O&M) is the longest phase of the cycle, it is necessary that the port is committed throughout this period, providing regular services and coordinating O&M activities, such as corrective and preventive maintenance, repairs and replacements. Its infrastructure requirements are less complex compared to the installation port.
Green port	It is the product of the long-term strategy aimed at sustainable and climate-friendly development of the port infrastructure. Environmentally responsible behaviour of all work structures, operations, equipaments and technologies. While many ports choose not to act beyond complying with existing environmental regulations, in many cases they have exercised their potential for addressing both social and environmental externalities.

Figure 3. Types of ports and their definitions. Source: adapted from [18,19,41,46–58].

Different port definitions have been provided to support offshore wind operations. In this article, a port industry as an offshore wind support port, from a point of view of all lifecycle phases, can be defined as an "industrial and logistical complex with features ranging from the manufacture of large components, their storage, assembly, and testing, as well as the support for the installation and operation and maintenance activities of offshore wind power plants, based on governance for sustainability of all its activities". Based on this concept, the factors that characterize this type of port will be described in the next sections.

4. Results

4.1. A Conceptual Model: Dimensions and Factors

A conceptual framework is presented in this section from the systematization of all the factors and characteristics identified during the SLR, with the main aim to characterize offshore wind support ports, including sustainability and organizational considerations, as important and necessary elements to make the port suitable. These elements were classified into six dimensions, as shown in Figure 4.



Figure 4. Conceptual framework of the dimensions for characterization of ports.

From the figure, the relationship between these dimensions can be seen. One is not superimposed on the other, but they are interconnected, complementing each other. The order from D1 to D6 means that in dimension D1, the factors are more tangible, technical, and quantifiable to analyze the port. But as the factors approach the D6 dimension, they appear to be more comprehensive and intangible, such as governance for sustainability, which is treated in a more global way.

The next topics describe, in detail, each of the factors collected in the SLR and the authors who cited them, divided into these 6 dimensions, pointing out the 71 systematized factors, in which there were 14 in physical characteristics, 17 in port layout, 7 in connectivity, 8 in port operation, 14 in performance optimization, and 11 in governance for sustainability. In the end, these factors will illustrate the proposed characterization framework.

4.1.1. Physical Characteristics

This dimension was defined to describe the physical structural port elements and availability of necessary equipment. In Table 2, the 14 factors are systematized, accompanied by the sources that cited them.

Factors	Definitions and Importance	References
Port's depth	Height between the seabed and the water depth of the port, suitable to accommodate vessels. Must be more than 10 m.	[8,11,12,15,18–20,41,43,48,50,56,57,59–78]
Quay length	It is the available linear extension and must be longer than the total length of the vessel, more than 200 m is acceptable.	[8,11,15,18,19,39,41,46,50,57,59–61,64– 68,73,75–80]
Quay width	It must be suitable for carrying components and for moving equipment. It must be wider than 70 m.	[18,64,69,70,73,80]
Surface loadbearing capacity	Measures the ability of the soil surface to support load, before failure occurs. Minimum 10 t/m ² .	[8,11,12,15,18–20,40,41,43,48,50,59,60,64– 72,77–79,81–84]
Appropriate docks and shipyards	Used for unloading, assembly, and construction of components and foundations. Suitability in surface, length, and depth.	[46,56,63,80–82,85–87]
Seabed suitability	Ability of the bed to accommodate vessels, in terms of soil conditions and composition, to bring stability to lifting vessels, as the jack-up vessel.	[8,11,12,41,59,62,64,66,68,70,78,82]
Availability of component handling equipment	Defined and available for loading, unloading, pre-assembly, and internal offsets.	[15,38,40,51,64,65,68,71,88–90]

Table 2. Factors classified by the physical characteristics of a port industry.

Factors	Definitions and Importance	References
Availability and capacity of suitable lifting cranes	They can be fixed, mobile, on tracks, gantry, and floating, with high load capacity, mainly for loading.	[8,11,12,15,19,20,39– 41,43,46,48,50,56,59,61,64– 70,72,73,75,76,78,80–83,85,86,90–98]
Availability of Ro–Ro and Lo–Lo	These are resources for integration to vessels that operate by rolling (Roll-on, Roll-off) and by lifting (Lift-on, Lift-off).	[11,15,41,46,59,65,66,68,78,88,99]
Availability of Self-Propelled Modular Transport Vehicles (SPMT)	Used to move the component in the port, with a superior flexibility and lower ground pressure compared to cranes.	[11,12,15,19,20,56,59,61,64,66,73,78,83,90,94]
Truck availability	For transporting smaller components, parts, and other resources.	[12,39,76,83,100]
Availability of floating pontoons	Two functions: manufacture of foundation by gravity and/or as a transport platform.	[22,59,61,82,88]
Appropriate port protection	Protection against cross currents, strong winds, storms, and waves. Loading bridges are essential.	[20,46,49,57,72,75,77,80,101]
Unrestricted air draft	Clear height with no overhead limitations, and no air current restrictions are required, especially for pre-assembly activity.	[15,19,48,49,56,57,59,60,64,66,67,73,75,77–80]

Table 2. Cont.

4.1.2. Port Layout

This dimension was chosen because the configuration of the port layout plays an important role in the efficiency of component installation operations. The response time can be reduced if the layout of the port is adequate; however, the opposite case will restrict all parts of the project [59]. Thus, in Table 3, the 17 factors are systematized.

Table 3. Factors classified into the port layout dimension.

Factors	Definitions and Importance	References
Component manufacturing facility availability	Large area for manufacturing installations of the main components. An average area of 15 hectares, depending on the number of factories and the space availability.	[11,12,14,15,19,20,40,46,48,50,56,57,59– 62,64–68,73,78,79,81,96,102,103]
Component storage area availability	Large area to supply the manufacture and assembly of components, to take advantage a time-limited weather window. It can be open or covered, using racks/fork-lifts. Minimum of 13 hectares, depending on the availability and dimensions of components.	[8,11,12,15,18,20,21,39– 41,46,48,49,52,56,57,59– 70,73,75,76,78,79,81,83,84,86,88– 90,94,96,104–116]
Spare parts storage area availability	Inventory storage for maintenance. May include components, handles, and tools.	[8,18,57,59,78,86,112,117–119]
Storage area for decommissioned components	Intended for storage, processing, and preparation for final disposal of decommissioned components.	[43,95,100,112]
Preparation/staging and assembly area availability	Intended to receive, prepare, pre-assemble, and load components before installation.	[11,12,14,15,18–21,39– 42,46,48,50,54,56,58,61,62,64,67,68,70,71,73– 79,81,86,90–92,94,96– 98,103,105,107,109,112,115,116,118,120–125]
Area for testing of components	In addition to assembly, an area for testing turbines is required before installation.	[14,70,75]
Proximity of storage and assembly to the quay	Storage and assembly must be adjacent to the quay, looking for minimal handling and movement.	[18,19,46,66,69,116]

Factors	Definitions and Importance	References
Dry dock area availability	Used to manufacture foundations, especially gravity or floating types.	[15,22,48,56,59-61,66,67,73,78,82]
Appropriate space for handling or movement	Unhindered port layout for handling and internal movement of components between areas.	[46,64,66,78]
Potential for expansion	It is essential to have an area in the port reserved for expansion, in case there is a need for future expansion, due to the prospects of industry growth and technological advances.	[11,18,41,68,101]
Office facilities and control centers	Includes offices, social rooms, rooms for maintenance activities, control, and human resources centers.	[11,18,41,46,56,57,64,66,67,69,70,75,78]
Loading and unloading area available	Large area, located on the quay or pier, with lifting cranes and ramps installed. The unloading area can also be on the land side.	[12,14,20,21,40,42,46,49,50,52,53,56,58,65, 66,77–80,89,93,94,96,97,106– 108,112,113,126,127]
Workshop area	For maintenance activities, repair of broken/defective components, inspections, cutting, and painting.	[11,19,41,56,61,63,64,67,73,78,109,111,112, 123,128]
Space to maneuver vessels	Enough space and depth to maneuver ships within the protected area of the port.	[46,101,108]
Heliport	Area for helicopter landing and technician transfer when the response time in O&M is critical.	[73,87,123]
Horizontal terminals	Organized as horizontally as possible, parallel to the pier for easy accessibility.	[18,67]
Onshore base green hydrogen and green ammonia	Base for the production, storage, and distribution of green hydrogen and ammonia to feed vessels, equipment, and factories. The space could also be used for fuel cell units.	[16,17,129,130]

Table 3. Cont.

The installation process requires the sequence of substructure, tower, nacelle, and blades and, therefore, storage at the port must follow this order, where support structures are closer to the pier and blades are stored further away, as they will be the last components installed [88]. In addition, each component must be transferred from the unloading area to the storage area, then to the preparation area, and finally to the loading area, to minimize the transportation cost [21].

4.1.3. Connectivity

The components, depending on size and weight, can be transported to the port by sea, road, and/or rail access [51]. In addition to multimodal connectivity, proximity to other key factors is also essential, hence the insertion of this dimension. Table 4 describes the seven factors.

Table 4. Factors classified into the connectivity dimension.

Factors	Definitions and Importance	References
Proximity between port and offshore wind farm	Key element for all phases because it significantly affects overall logistics strategy, time, and cost, being especially critical for O&M.	[11,12,14,18,19,21,39– 43,46,56,57,59,63,64,67,69,72,73,77– 79,81,82,85,86,91,94,96,97,105,107,110,112– 114,117,118,121,123,126,131–133]
Proximity to the key component suppliers	Having regional suppliers of key components near or within the port area is an advantage.	[11,15,18,40,41,56,58,61,62,68,69,92,126,134]

Factors	Definitions and Importance	References
Proximity to heliports	Support for the O&M phase, when response time is critical.	[11,18,41,46,56,59,61,65,79,85,87,132]
Proximity to airports	To receive priority resources, with agility.	[18,19,46,56,61,64]
Proximity and access to road and rail networks	There must be road and rail connections for transporting and loading/unloading components and materials.	[8,11,15,18,19,41,46,48,51,54,56,59,61,64,67– 70,72,73,75,77,78,83,90,95,100,111,112,116, 133,134]
Appropriate maritime access	It must allow the transit of specialized vessels.	[12,19,20,46,48,49,51,54,56,73,80,90,98,108, 112,116,120,126,132,135]
Wide navigation channels	Must be clear and direct, with a wide navigable entrance and adequate width, unrestricted horizontal and vertical clearance.	[12,46,48,56,57,62,64,66,72,73,75,78]

Table 4. Cont.

4.1.4. Port Operation

Port operation for this type of complex activity requires needs on the part of operators as well as the port management, justifying the choice of this dimension as part of the result. Table 5 describes the eight factors.

Factors	Definitions and Importance	References
Port operationalization, 24/7	The port must operate 24 h/day, 7 days/week, 365 days/year. In a marine environment, a maximum of 12 h/day.	[18,56,57,59,64–66,73,75,77,78,121]
Specialized workforce	Availability of specialized workers, mainly local, such as engineers, electricians, mechanics, and machine operators.	[15,18,46,58,59,73,75,78,79,86]
Sharing information	It is essential to have an installation schedule combined with the sharing of information about the port's storage capacity, weather forecast, and availability of vessels.	[12,38,59,65,84,103,105,107,126,129]
Team and supplier management	To avoid stockout situations. These skills are needed to ensure the efficient operation of the supply chain and logistics.	[40,49,77]
International port operation compliance	Ensure that cargo can be transported internationally, requiring an ISPS (International Ship and Port Facility Security) compliant port.	[15]
Occupational health and safety measures	Measures and strict requirements must be implemented to avoid riskiness and accidents on land and/or sea. Establish occupational health programs to protect everyone.	[58,59,61,65,136]
Facilities for first aid at the port	Provide emergency services and first aid facilities. The port must be sufficiently close to hospitals.	[56,136]
Training courses and research center	Port location with training courses and research in centers of excellence. Promote access to high-quality training and research expertise at the port facility or through courses and training elsewhere or centers of excellence near the port.	[18,75,102]

Table 5. Factors classified into the port operation dimension.

4.1.5. Performance Optimization

This dimension is essential because, on the one hand, there is a need to assign port resources very early in the lifecycle phase of the offshore wind farm and, on the other hand, an inadequate dimensioning of the ports can drastically interfere in efficient installations and incur additional costs. For that reason, each process in the performance is crucial to minimize costs, logistical inefficiencies, and waiting times [108,137]. The 14 factors are described in Table 6.

Table 6. Factors classified into the performance optimization dimension.

Factors	Definitions and Importance	References	
Minimum port–farm distance	Impacts on port costs because it reduces transfer time, uses less fuel, expands the climate window, and decreases LCOE.	[8,11,18,41,42,52,64,66,69,76,78,85,87,88,97, 100,110,122,123,125,127,133,138]	
Port storage cost	It is included in the logistical costs. Corresponds to port fees for the storage of parts and facilities.	[90,118]	
Port labor cost	Multiply the average working days required by the fixed daily work rate.	[65,66,87,118]	
Annual rate	Paid to local authorities for the use of port infrastructure, berthing at the quay, and use of cranes.	[61,118]	
Learning rate	It shows how agile the operations are. If the rate is high, then the shipping and installation cost is low.	[97]	
Transport of complete structures	Included in the port cost. It aims to minimize offshore work and maximize pre-assembly onshore, simplifying operations, time, and costs.	[42,43,110,122,123,125,139]	
Inventory control and targets	Port inventory control that synchronizes storage, production, and lead times to maintain capacity, avoid waiting times, and ensure supply.	[20,38,79,84,88,106,115,116,137,140]	
Optimization of the required flow of inputs and outputs to components	It is essential to optimize the cycles or inflow and outflow of the turbines, to ensure replenishment and deliver components at a defined frequency, avoiding delays, additional costs, and low performance.	[137,140]	
Potential activities and use conflicts management	Ports host a variety of activities other than offshore wind, which can conflict with each other directly or inadvertently, positively or negatively.	[78]	
Loading time at port	Included in the port cost. This time depends on the configuration, how many turbines will be loaded, and the number of lifts.	[49,52,78,104,106,110,115,116,119,137,139]	
Transport time from the port to the farm	This transport time depends on the number of turbines, vessel speed, and port-to-site distance.	[52,66,77,78,104,106,115,116,119]	
Decommissioning cost	Means the transportation cost of the return and processing at the port of decommissioned components.	[110]	
Choice of vessel type	This is a crucial decision when planning the installation, as well as choosing the loading mode and the base port.	[61,110,139]	
Towing speed	Included in the port cost. Corresponds to the speed of the tugboat when transiting components from the port to the site. The higher the speed, the higher the performance.	[110]	

4.1.6. Governance for Sustainability

Governance of ports is a very important issue that is often overlooked in many studies that only consider logistics characteristics. Therefore, this dimension was included because, as demonstrated by the literature analyzed in Table 7, sustainable development in port regions is necessary and is characterized by a complex decision-making process involving environmental, social, and economic issues [141].

 Table 7. Factors classified in the governance for sustainability dimension.

Factors	Definitions and Importance	References
Storage and use of renewable energy	Use the port to store excess energy obtained from wind farms, using hydrogen. The port would function as a generator and energy consumer, being able to achieve 25% less emissions.	[16,57,129,142–144]
Use of OPS system	Port power supply, called onshore power supply (OPS). Allows the shutdown of auxiliary engines of vessels while they are berthed. Reduces adverse environmental effects, generating energy from renewable sources.	[129,130,142,145]
Monitoring of port environmental impacts	Fundamental in all phases of construction and operation, as dredging activities can induce aquatic impacts, and port operations can produce sewage and solid waste, leakage of harmful materials, and oil pollution.	[136,144]
Investment in green equipment	Equipment is less likely to cause pollution, such as engines from renewable sources and electric cranes.	[47,57,136,144,145]
Sustainability practices in the port operations and its entire chain	The entire supply chain at all stages must be concerned with following environmental criteria and sustainable business practices with equal weight to environmental, economic, and social decisions. It should include the efficient use of resources, quality of the natural port environment, waste management, waste reduction, and encouragement to reuse and recycle materials.	[47,57,64,65,144,146–148]
Clean and efficient shipping, transport, and logistics	The ship-building industry must implement strict liquid and solid waste policies and use electric power and alternative fuel engines. The "Clean Truck Program" consists of a ban on trucks manufactured in the port before 1989.	[144,145,148]
Fuel cells for operations and vessels	Green energy strategy for a system composed of hydrogen fuel cells, electromechanical cells, and offshore wind turbines. It provides a reduction in CO_2 , NOx, and CO emissions from vessels, in addition to the economic impact of the application of this technology.	[16,17]
Inclusion of trees in the port area	It is a landscape inclusion that includes trees, with the objective of absorbing noise and reducing pollution, reducing the contributions of greenhouse gases and other emissions.	[144,147]
Monitoring composite sustainability indexes	Use of composite sustainability indexes, encompassing the three dimensions: economic (regional GDP, employment and unemployment rates, and employment generation mainly from manufacturing and O&M facilities), social (gender differences, life expectancy, rate of participation in education and training, and benefits to the local community), and environmental (air pollution and average annual exposure). Thus, it is possible to measure the positive impact of the sustainable development of the ports in a composite way.	
Development and maintenance of highways and railways	The port as a driver in the development of highways and railways, mainly those connected to it.	[129,135]
Link with population centers	Connection and proximity of the farms to population centers is essential, because it will allow less tension in terms of accessibility and logistics of the network, promoting social development. This is primarily a "connectivity" issue but has also been placed here due to its sustainability implications.	

Also, it is important to realize that the environmental impact of ports and the problems caused come from three main sources: the port activities, the activities of the ships that dock, and emissions from the intermodal transport chains that serve the interior of the port [129]. Thus, the negative environmental impact is a reality [142]. Hence, offshore wind support ports also contain these characterization factors, because minimizing negative environmental impact is necessary in order not to negate some of the positive environmental benefits gained by the transition to offshore wind energy. Equally, the economic and social benefits that an offshore wind port brings should be managed in a sustainable manner. Table 7 describes 11 factors covered, which show how the green port could be operated and managed.

4.2. A Proposal Framework: Factors for Characterization of a Green Port Industry

This section consists of validating the developed framework, through the results of the application of a questionnaire answered by experts in the field of ports for offshore wind. The following were considered as experts: authors with the highest number of publications (greater than or equal to three), the first two authors of the technical reports analyzed, speakers at international events on ports, port managers of the main ports that support the offshore wind sector, and experts in the study of the offshore wind sector.

Some of these respondents noted that the analysis of the exposed factors depends on the type of port that is intended to be analyzed, what the support is, and at what phase of the lifecycle, as there are many types, with different scopes of work and different requirements to be attended, consequently.

However, the objective of this study was to characterize a green port industry to support the offshore wind sector as a port capable of providing the necessary support, from the manufacture of components within the port area, through the subsequent support for installation, operation, and maintenance, and even its decommissioning. Because of this, the research systematized all factors found in the publications, not focusing only on factors corresponding to an installation port or an O&M port, for example, nor on supporting a specific lifecycle phase.

Regarding the dimension "physical characteristics", no factor needed to be removed, as most factors obtained 100% agreement. The factor "appropriate docks and shipyards" was disagreed on by some, as seen, but it was not removed due to its importance in the manufacture of foundations, mainly. Therefore, they are in fact not essential in the installation phase, as commented by experts, but they are essential in the manufacturing phase of fixed and floating foundations, as well as being a support point for unloading and assembly when necessary [56,85,86]. Likewise, the factor "availability of floating pontoons" was defended, as it supports the same function of building foundations, mainly of the gravity type [82].

The other factor also commented on with disagreements was "seabed suitability", but this factor was not removed, as it is important not only for the location of the farm, as justified by the specialist, but mainly to know the conditions and capacity of the seabed of the port to dock and accommodate certain vessels in order to stabilize them at the time of lifting for loading, as occurs with the jack-up vessel [11,12].

Regarding the suggestion of new factors, according to the experts, the inclusion of the factor "capacity and availability of energy substations" was considered, since it is really an important factor to be analyzed with a view to supplying sufficient energy resources for the operation of the port and execution of manufacturing, assembly, and loading, among others. The factor "wide evolution basins" was not considered as a new factor, as it is already included in the factor "wide navigation channels", in the dimension "connectivity". Likewise, the suggested factor "differentiation between SOVs and CTVs" was also not included, as it was already included in the factor "choice of vessel type", in the dimension "performance optimization".

When analyzing the results for the dimension "port layout", no factor was removed or added. The first factor that obtained disagreement was "dry dock area availability", with the justification that they are not essential in the port, but looking at an area suitable for manufacturing gravity and floating support structures (foundations), as well as their maintenance and repair [22,59], becomes an important factor to be analyzed, depending on the strategy adopted by the developer and stakeholders.

The "heliport" factor was also considered not essential for installation or manufacturing. However, it was not removed, as it is an extremely important factor for maintenance activities with a critical response time for resolution and high complexity [122], which can also occur in a port industry. The third and last factor with a percentage of disagreement was "onshore base for storage of green hydrogen and green ammonia", with the justification that it is not essential for the port, but for the green energy transition.

However, the port, according to good practices seen in countries such as Scotland, can serve as a viable and strategic base not only for storage, but also for the production, distribution, and export of green hydrogen, for example [16,17,129,130].

For the "connectivity" dimension, the results showed that the "proximity to heliports" factor remained for cases where there is no heliport in the port area. However, the "proximity to airports" factor, the second that had disagreements, was removed, as experts believe it is unnecessary in view of the other forms of existing connectivity. The other factors obtained 100% agreement and there was no addition of new factors for this dimension.

For the dimension "port operations", the factors remained unchanged. There were no justified disagreements or suggestions for new factors. Regarding the "performance optimization" dimension, the main result was the need to clarify the dimension and what it is about. The disagreement or agreement in parts was indicated by thinking that the dimension should only consider port assessment factors or those that directly involve port operations.

However, in fact, the dimension seeks to bring important decision factors regarding the interface between the port and farm, costs and fees that involve both sides directly and indirectly. Each of the factors considered is defined and justified in Tables 4 and 5. All have a direct or indirect influence on the port cost, such as towing speed, decommissioning cost, the minimum port-to-farm distance, and the choice of vessel type, as it influences the type of pre-assembly, time, and mode of loading operations, among other essential relationships. With this, the dimension, in its latest version, is now called "port–farm performance optimization".

Thus, none of the factors were excluded, the respondents' comments did not include plausible justifications for the removal of any factor. Regarding the suggestions for new factors, they were also not added, since the possible factor "rapid adaptation to bad weather conditions" is already considered in the factor "optimization of the required flow of inputs and outputs to components" in this same dimension, as it concerns precisely predicting production and demand and knowing how to deal in an optimized way in adapting to bad weather conditions through stocks and replenishment cycles, to avoid delays, additional costs, and poor performance [137].

In the "governance for sustainability" dimension, there was a comment about these sustainability factors and decisions still being in the research and discussion phase to be clearly determined, as achieving and maintaining greater sustainability is an important factor, but the location of the facility and the capabilities are even more urgent in practice. Through the results obtained throughout the study, this understanding is clear, but the novelty of the research was to fill precisely the existing gap on the discussion of sustainability factors and how this governance could occur in the port environment and which paths could be followed.

Thus, some factors disagreed. The factor "fuel cells for operations and vessels" was not removed, as it is known that the use of combustion engines can still be more economical and efficient, but the use of fuel cells will be a viable and extremely necessary option in the near future, as a green energy strategy that can significantly reduce CO₂, NOx, and CO emissions from ships, in addition to the economic impact of applying green energy technologies [16,17].

The factor "monitoring composite sustainability indexes" was also not removed, as in addition to the disagreement not having been justified, this factor brings together a number of important factors for achieving sustainability and green port policy, which seeks to encompass the three dimensions' analysis: economic, environmental, and social [103,141].

The factor "inclusion of trees in the port area" was removed, as the experts' justification was plausible: there are other more important factors with better academic basis to be

considered, regarding the mitigation of environmental impacts and promotion of green ports, which go beyond this concern for the port's landscape.

In addition, the insertion of a new factor "presence of recycling companies" was suggested, but it was not added to the proposed model, as it is already being considered in the factor "sustainability practices in the port operations and its entire chain" in the same dimension.

Therefore, some factors were added and removed after the validation and discussions step, as explained. Thus, the proposed new model is illustrated in Figure 5. There were 71 factors, where 2 factors were excluded (proximity to airports, in the connectivity dimension, and inclusion of trees in the port area, in the governance for sustainability dimension) and 1 factor was included (capacity and availability of power substations, in the physical characteristics dimension).



Figure 5. Proposal characterization framework.

4.3. Case Studies on Existing Ports: Strengtheing the Proposed Framework

This section consists of strengthening the proposed framework through comparison and analysis of existing ports to validate some factors and characteristics already identified, especially the quantitative ones. The studied ports were (1) the port of Esbjerg in Denmark, (2) the port of Hull in the United Kingdom, and (3) the port of Bremerhaven, Germany. The port of Esbjerg was selected as the reference port in support of the offshore wind sector and helped to implement the largest installed capacity of offshore wind farms. The port of Hull was studied due to its experience of including the blade industry in the port, having preparation areas for the installation of offshore wind farms in the UK, as well as being considered a green port and industry port. The port of Bremerhaven was chosen for its expertise and experience with the offshore wind industrial cluster.

Table 8 shows the three ports in terms of physical characteristic factors, port layout, connectivity, as well as some governance aspects for sustainability, which demonstrate the main aspects of a green port industry to support the offshore wind sector, for the manufacturing operations, installation, and operation and maintenance of plants, also linked to the implementation of green port strategies.

Factors	Bremerhaven	Esbjerg	Hull
Port type	Port industry, as well as a base for assembly, installation, and O&M.	Base port for installation and O&M.	Port industry, as well as a base for installation and O&M.
Operations/ Activities	Manufacture and assembly of foundation, tower, and nacelle, manufacture of blades, as well as storage, transport with suitable load capacity, and heavy forklifts. Also included the export of components.	Manufacturing, storage, assembly, testing, loading, unloading, and transport.	Manufacturing, storage, preparation, loading, and transport.
Supporting farms	25	55	20
Area (offshore wind)	More than 25 ha (OTB); 10 ha (ABC); 25 ha (containers 1)	450 ha (total)	58 ha (Alexandra Dock)
Port's depth	14.1 m (OTB); 10.5–11 m (ABC); 12.5–15.5 m (containers 1)	9.4–10.5 m	8.3 m (Alexandra Dock)
Quay length	500 m (OTB); 900 m (ABC); 450 m (containers)	14 km	4.082 m (Alexandra Dock)
Surface loadbearing capacity	10–50 t/m ² (OTB); 20 t/m ² (ABC)	15–30 t/m ²	20 t/m ² (Alexandra Dock)
Crane capacity	30–400 t	308–448 t	180–420 t
Storage area	Approx. 100,000 m ²	Approx. 1,000,000 m ²	70,000 m ² —Covered 650,000 m ² —Open
Manufacturing area	Yes (blades, foundations, towers, cables)	Yes (MHI Vestas)	Yes (Siemens Gamesa, 40,000 m ²)
Assembly/ preparation area	Yes	Yes	Yes
Testing area	Yes (blades and nacelles)	Yes	No
Heliport	No	Yes	No
Dry dock	Yes	Yes	Yes
Expansion area	Approx. 200 ha	Approx. 100 ha	Approx. 183 ha

Table 8. Characterization factors for the ports of Bremerhaven, Esbjerg, and Hull.

Factors	Bremerhaven	Esbjerg	Hull
Horizontal terminals	Yes	Yes	Yes
Proximity and access to road and rail networks	Yes	Yes	Yes
Proximity and access to airports/ heliports	Yes (58 km from Bremen International Airport)	Yes (8 km from Esbjerg Airport and heliport)	Yes (28 km from Humberside Airport, which houses the second largest heliport in the country)
Wide navigation channels	11–12.2 m (depth); 498 m (width)	10.3–11.6 m (depth); 200 m (width)	7.9 m (depth); 167.7 m (width)
Port operation 24/7	Yes	Yes	Yes
Specialized workforce/ jobs	Yes (many training centers and generation of 3000 jobs)	Yes (generation of 10,000 jobs in the offshore wind and oil and gas sector, with specialties of blacksmiths, electricians, welders, and engineers)	Yes (12,000 overall and over 3500 in the offshore wind chain)
Training centers/ universities	Fraunhofer (IWES); Deutsche WindGuard GmbH; k-wind: University of Bremerhaven; ISL Institute for Shipping Economics and Logistics; Wind Energy Agency (WAB)	Offshore Academy; Energy Innovation Cluster	HETA; HOTA; Hull Training; HFR Solutions; Universidade de Hull
Distance to the farm	Up to about 320 km	Up to about 600 km	Up to about 300 km (12 h of navigation to wind farms of the third round)
Governance for sustainability and benefits for the local community	PERS certification; ESI Index; recycling system; use of terrestrial energy with renewable energy sources; air quality indicators; CO ₂ footprint analysis; sediment management; defined compensation and protection measures; city development; highway development; incentive to local companies; employment and income; measures to disseminate to the public (observation facilities, guided tours, and events).	Yellow containers for garbage collection; recycling station; effective waste disposal system; port waste treatment plan; city development; development of rail and road access; employment and income; opportunity for young people.	Biomass handling facility; Energy Works Project; Chowder Ness; noise barrier; opportunity for young people and women; Humber Coastal Half Marathon.

Table 8. Cont.

Source: adapted from [149–179].

From this systematized table, it is possible to compare with the minimum or ideal values established in Section 4.2 to conclude if what is discussed in the literature and technical reports can be proven in the reality of existing industrial ports. For examples, the verification and comparison for each requirement follow:

(a) The surface loadbearing capacity was established with at least 10 t/m^2 . In the studied ports, it appears that this requirement is being met, as shown in Table 8.

(b) The capacity of the cranes was observed with the ideal of 400 t to 1000 t. In the studied ports, all identified cranes reached 400 t, which, added to the capacity of other cranes, working together, can exceed 1000 t.

(c) The quay length established as the minimum acceptable was 200–300 m, as an average value found, considering the technological and dimensional advances of the vessels

and components. The case study ports exceeded this limit in their terminals and wharves, which proved compliance with this requirement and full conditions for berthing vessels.

(d) The minimum depth of the port industry was established at 10 m, as a minimum acceptable average for suitability for receiving vessels. The case studies showed that two ports were following this requirement, Esbjerg and Bremerhaven, while the port of Hull was at 8.3 m, the smallest depth among the three.

(e) The significant use of SPMTs and Ro–Ro ramps was proven in the literature and in the conceptual framework. It was also identified as good practice in the case studies, since all ports make these tools available in the execution of their operations.

(f) Layout areas, as noted in the systematic literature review, vary greatly depending on the availability of area in each port. However, some values perceived as usual were 20 to 40 ha for the storage and assembly area, up to 40 ha for manufacturing facilities, and an average area of 4000 m² for offices and control centers. In the case studies, this information was not available for the most part, but the storage areas found were within compliance, if compared to this information.

(g) Factors established as important in a port industry, such as expansion area, testing area, presence of dry docks, heliport in the port area, and horizontal terminals, were verified in the case studies and were compliant, showing that they are also able to use these good practices.

(h) Another requirement that was also met by the three ports studied was proximity and access to road and rail networks, as an essential factor for unloading raw materials for manufacturing and small components.

(i) The port operation in the 24/7 regime was also a factor attended by all the studied ports, as a good practice, mainly in the operation and maintenance phase. In addition, the specialized workforce, training centers, and partnerships with research and development centers were proven as important factors and were also part of the practices established in the three ports of the case studies.

(j) The distance between the port and farm has an average requirement of 150 km to support O&M activities and an average of up to 400 km to support plant installation activities, as verified in Section 4.2. The studied ports complied with these requirements mainly regarding installation activities, with the port of Esbjerg being the one with the greatest reach in terms of service to the farms.

(k) All ports in the case study can also be considered green ports, as a good practice identified in the SLR, meeting numerous governance factors for sustainability and the consequent benefits to the local community.

5. Discussion

In addition to the direct relationships that the factors establish within each dimension by affinity, it is also possible to extract from the framework factor relationships between different dimensions, such as the following: the close relationship between the specialized workforce and the generation of employment and income; the need for access to roads and railways and their consequent regional development; the need to install factories with the disposal of suppliers close to or within the port area; the depth of the port and the close connection with the suitability of the docks, seabed, and the choice of vessels; the areas for handling and movement with the respective availability and capacity of handling, lifting, and transport equipment; the existence of training centers with an increase in the learning rate; the relationship of the information sharing factor with the control factors and inventory targets, management of suppliers and teams, as well as the promotion of a sustainable chain, among other interconnections.

It was also noted that the systematization globally sought all the factors that can characterize a support port for the offshore wind sector, regardless of the phase it serves or the target activity it is willing to offer, designing an ideal port that encompasses industry at all stages and is sustainable, although more factors have been found for the manufacturing and installation phases. In addition, although the factors have been systematized qualitatively, some factors have quantifiable requirements, mainly the factors of the dimensions "physical characteristics" and "layout". It was also proven by the case studies described in the last section.

Depending on the conditions of the port area, meeting one requirement to the detriment of another will result in a better or worse cost–benefit, which proves, once again, the close relationship between the factors and the impact of each one on the cost, performance, and consequently, on the decisions. For example, if there is an area to build a port with a shallow seabed depth, there will be the cost of dredging to reach the minimum height of 10 m. If there is another area with an adequate depth, but without a sufficient coastline length and width for berths and quays, there would be this cost for the potential increase in available coastal area. This is a multicriteria decision based on the conditions of each port area analyzed.

Therefore, this research can be seen as the advancement and implementation of a more holistic and systematized operational and sustainable analysis, in which 71 characterization factors were identified and classified into 6 dimensions, thus providing a novel, detailed, 6-dimension systematic analysis of the literature pertaining to ports that support the offshore wind sector. In addition to the structured dimensions and factors validated with experts, the novelty of this article is also the new port concept presented, which combines not only installation and operation support activities, but also component manufacturing activities as an industrial cluster, in a port environment with green technologies and management.

6. Conclusions

The accelerated technological development of the offshore wind turbine requires that the traditional function of the port be expanded from a logistical port to a port with manufacturing, assembly, and testing facilities. Also, the concern for the sustainability of the planet requires that ports adopt strategies involving the concept of a green port and actions of governance for this sustainability in all the operations and management. Thus, new requirements, factors, and characteristics are needed to have suitable ports to support the offshore wind sector.

Because of the above, this study was conducted based on the following research question: which factors should be considered to characterize an offshore wind support port? And this article was achieved with the main objective being to identify the factors that characterize a suitable port to support the offshore wind sector. This was reached using SLR and the conceptual framework proposal as a result, with 71 characterization factors classified into 6 dimensions, not only related to technical feasibility, but also to infrastructure and operational factors, optimization and organizational, and especially sustainability elements, which are able to assess what a port could have.

The proposed framework was strengthened by case studies on existing ports, which identified a list with the minimum requirements of a port capable of this level of support, as well as described other good international practices at these specific ports, which also contributed to and reinforced the characterization of a green port industry.

Therefore, the identification of these factors contributes not only to the state-of-theart, but also to the offshore wind energy sector and the port sector, because they can promote less risk in manufacturing, installation, and O&M operations, as well as influence the increase in competitiveness of both ports and project developers. Linked to this, the research can promote benefits for other port economic activities, as well as for regional development and local communities. There is a vision not only of the importance of ports for offshore wind, but also the benefits of offshore wind as a competitive advantage for ports.

Thus, the choice, adequacy of ports, and projection of new port infrastructure will be a focal point of analysis for the projects, the developers, the port authorities, and other stakeholders involved, as well as for formation and discussion of public policies, due to the accelerated and continuous evolution of this industry and this technology. Also, the results of this research are essential to decision-makers about how to invest in adequacy of ports or construction of new ports to support this sector, being able to take this study as a guide of factors and characteristics to be analyzed and prioritized.

The results highlight the importance of this research, although it faced some limitations, mainly about the difficulty of keeping in touch with experts and the sample size obtained in the study.

For future research, it is recommended to analyze what minimum or ideal requirements are necessary so that it is suitable for the dimensions of the components of the next generations of turbines. Furthermore, how can the identified factors be analyzed and classified according to the level of criticality, so that the priority factors are listed? What multicriteria decision models can be developed from the identification of these factors? What mathematical models can be developed to evaluate the performance of these ports in supporting offshore wind? What paths should decision-makers, port authorities, and governments follow to secure investments in ports? How could a technical, economic, and environmental feasibility study be conducted for this type of port?

Author Contributions: Conceptualization, M.G. (Monalisa Godeiro) and M.G. (Mario González); methodology, M.G. (Mario González) and M.G. (Monalisa Godeiro); validation, M.G. (Mario González), D.J., N.A., G.N., A.S., D.M., R.V., L.N., M.A. and J.T.; formal analysis, M.G. (Monalisa Godeiro); research, M.G. (Monalisa Godeiro); data curation, M.G. (Monalisa Godeiro) and M.G. (Mario González); writing—preparation of the original draft, M.G. (Monalisa Godeiro); writing—review and editing, M.G. (Mario González), D.J. and N.A.; supervision, M.G. (Mario González); project administration, M.G. (Mario González). All authors have read and agreed to the published version of the manuscript.

Funding: This study was supported by Petrobras and the Federal University of Rio Grande do Norte (UFRN), through ANEEL's R&D—Electric Sector (PD-00553-0045/2016); the Coordination for the Improvement of Higher Education Personnel of Brazil (CAPES); the Ministry of Science, Technology, and Innovation (MCTI)/Secretariat for Technological Development and Innovation (SETEC)—TED No. 11355260/2023; the National Council for Scientific and Technological Development (CNPq)—Process No. 308665/2020-1 and No. 406746/2022-2—of Brazil; the government of the state of Rio Grande do Norte, through SIN and SEDEC—Agreement No. 002/2023.

Data Availability Statement: The original contributions presented in the study are included in the article. Further inquiries can be directed to the corresponding author.

Acknowledgments: We would like to thank all survey questionnaire respondents, as well as the Federal University of Rio Grande do Norte for their support and infrastructure and the offshore wind experts and researchers from the Creation Research Group for all discussions and knowledge.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- González, M.O.A.; Gonçalves, J.S.; Vasconcelos, R.M. Sustainable development: Case study in the implementation of renewable energy in Brazil. J. Clean. Prod. 2017, 142, 461–475. [CrossRef]
- 2. GWEC. Global Wind Report 2023; GWEC: Brussels, Belgium, 2023.
- 3. IRENA. Global Renewables Outlook: Energy Transformation 2050; IRENA: Abu Dhabi, United Arab Emirates, 2020.
- 4. World Bank. Going Global-Expanding Offshore Wind to Emerging Markets; World Bank: Washington, DC, USA, 2019.
- 5. GWEC. Global Offshore Wind: Annual Market Report 2020; GWEC: Brussels, Belgium, 2020.
- 6. IEA. Offshore Wind Outlook 2019: World Energy Outlook Special Report; IEA: Paris, France, 2019.
- 7. Falani, S.Y.A.; González, M.O.A.; Barreto, F.M.; Toledo, J.C.; Torkomian, A.L.V. Trends in the technological development of wind energy generation. *Int. J. Technol. Manag. Sustain. Dev.* **2020**, *19*, 43–68. [CrossRef]
- 8. European Union. FOWPI—Coastal Aspects and Port Requirements; European Union: New Delhi, India, 2018.
- 9. Siemens Gamesa Renewable Energy. SG 14-222 DD Offshore Wind Turbine 2020. Available online: https://www.siemensgamesa. com/en-int/products-and-services/offshore/wind-turbine-sg-14-222-dd (accessed on 2 December 2020).
- Goldwind. World's First 16MW Offshore Wind Turbine Installed Offshore China. In: Offshore Eng. Available online: https://www.oedigital.com/news/506195-world-s-first-16mw-offshore-wind-turbine-installed-offshore-china (accessed on 27 July 2023).
- Akbari, N.; Irawan, C.A.; Jones, D.F.; Menachof, D. A multi-criteria port suitability assessment for developments in the offshore wind industry. *Renew. Energy* 2017, 102, 118–133. [CrossRef]

- 12. Thomsen, K.E. Offshore Wind a Comprehensive Guide to Successful Offshore Wind Farm Installation, 2nd ed.; Academic Press: Cambridge, MA, USA, 2014. [CrossRef]
- 13. Poulsen, T. Logistics in Offshore Wind. Ph.D. Thesis, Aalborg University, Aalborg, Denmark, 2018. [CrossRef]
- 14. Poulsen, T.; Lema, R. Is the supply chain ready for the green transformation? The case of offshore wind logistics. *Renew. Sustain. Energy Rev.* **2017**, *73*, 758–771. [CrossRef]
- 15. Fowind. Fowind—Supply Chain, Port Infrastructure and Logistic Study for Offshore Wind Farm Development in Gujarat and Tamil Nadu; GWEC: Brussels, Belgium, 2016.
- 16. Seddiek, I.S. Application of renewable energy technologies for eco-friendly sea ports. *Ships Offshore Struct.* **2020**, *15*, 953–962. [CrossRef]
- 17. Sadek, I.; Elgohary, M. Assessment of renewable energy supply for green ports with a case study. *Environ. Sci. Pollut. Res.* 2020, 27, 5547–5558. [CrossRef]
- Jäppelt, U.; Carstensen, N.; Schnabel, F. Hafenkonzepte für die Offshore-Windindustrie: Anforderungen an Offshore-Häfen. Beton- Stahlbetonbau 2012, 107, 33–38. [CrossRef]
- Cooper, B.D.; Marrone, J.F. Port Requirements to Support Offshore Wind Development in North America. In *Ports 2013: Success Through Diversification*; American Society of Civil Engineers: Reston, VA, USA, 2013; Volume 2013, pp. 1473–1482. [CrossRef]
- Matson, C.B. Practical Design of a Wind Turbine Logistics Port. In *Ports*; American Society of Civil Engineers: Reston, VA, USA, 2016; Volume 2016, pp. 303–309.
- 21. Irawan, C.A.; Song, X.; Jones, D.; Akbari, N. Layout optimisation for an installation port of an offshore wind farm. *Eur. J. Oper. Res.* **2017**, *259*, 67–83. [CrossRef]
- 22. Jiang, Z. Installation of offshore wind turbines: A technical review. Renew. Sustain. Energy Rev. 2021, 139, 1105760. [CrossRef]
- 23. Demo, P. Metodologia do Conhecimento Científico, 1st ed.; Atlas: São Paulo, Brazil, 2000.
- 24. Creswell, J.W. *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches,* 3rd ed.; SAGE Publications, Inc.: Thousand Oaks, CA, USA, 2009.
- 25. Marconi, M.; Lakatos, E. Fundamentos de Metodologia Científica, 5th ed.; Atlas: São Paulo, Brazil, 2003. [CrossRef]
- 26. Yin, R.K. Estudo de Caso: Planejamento e Métodos, 2nd ed.; Bookman: Porto Alegre, Brazil, 2001; Volume 369.
- 27. Voss, C.; Tsikriktsis, N.; Frohlich, M. Case research in operations management. *Int. J. Oper. Prod. Manag.* 2002, 22, 195–219. [CrossRef]
- González, M.O.A.; Toledo, J.C. A integração do cliente no processo de desenvolvimento de produto: Revisão bibliográfica sistemática e temas para pesquisa. *Produção* 2012, 22, 14–26. [CrossRef]
- 29. Jesson, J.; Matheson, L.; Lacey, F.M. *Doing Your Systematic Review—Traditional and Systematic Techniques*, 1st ed.; Sage Publications: London, UK, 2011.
- 30. Kitchenham, B.; Pearl Brereton, O.; Budgen, D.; Turner, M.; Bailey, J.; Linkman, S. Systematic literature reviews in software engineering—A systematic literature review. *Inf. Softw. Technol.* **2009**, *51*, 7–15. [CrossRef]
- Cook, D.J.; Mulrow, C.D.; Haynes, R.B. Systematic Reviews: Synthesis of Best Evidence for Clinical Decisions. Ann. Intern. Med. 1997, 126, 376–380. [CrossRef]
- 32. Pai, M.; McCulloch, M.; Gorman, J.D.; Pai, N.; Enanoria, W.; Kennedy, G.; Tharyan, P.; Colford, J.M., Jr. Systematic reviews and meta-analyses: An illustrated, step-by-step guide. *Natl. Med. J. India* **2004**, *17*, 10.
- 33. Webster, J.; Watson, R.T. Analyzing the Past to Prepare for the Future: Writing A Literature Review. *MIS Q.* 2002, 26, 13–23. [CrossRef]
- 34. BVG Associates. Oil and Gas 'Seize the Opportunity' Guides: Offshore Wind; Scottish Enterprise: Glasgow, Scotland, 2018. [CrossRef]
- Creation. Cadeia de Valor da Indústria Eólica Offshore. 2019. Available online: https://abeeolica.org.br/wp-content/uploads/ 2023/01/Sumario-Executivo.pdf (accessed on 15 February 2020).
- 36. DTOcean. Methodology Report and Logistic Model Flow Charts; France Energies Marines: Plouzané, France, 2014; Volume 1.
- 37. González, M.O.A.; Santiso, A.M.; de Melo, D.C.; de Vasconcelos, R.M. Regulation for offshore wind power development in Brazil. *Energy Policy* **2020**, *145*, 15. [CrossRef]
- Ait-Alla, A.; Quandt, M.; Beinke, T.; Freitag, M. Improving the decision-making process during the installation process of offshore wind farms by means of information sharing. In Proceedings of the Twenty-Sixth International Ocean and Polar Engineering Conference, Rhodes, Greece, 26 June–2 July 2016; pp. 144–150.
- Vis, I.F.A.; Ursavas, E. Assessment approaches to logistics for offshore wind energy installation. *Sustain. Energy Technol. Assess.* 2016, 14, 80–91. [CrossRef]
- Vojdani, N.; Lootz, F. Designing supply chain networks for the offshore wind energy industry. Int. J. Bus. Perform. Supply Chain. Model. 2012, 4, 271–284. [CrossRef]
- Akbari, N.; Irawan, C.A.; Jones, D.F.; Menachof, D. The role of ports in the offshore wind power industry. In *Port Management: Cases in Port Geography, Operations and Policy*; Pettit, S., Beresford, A., Eds.; Kogan Page Publisher: London, UK, 2017. Available online: https://www.koganpage.com/product/port-management-9780749474324 (accessed on 3 September 2020).
- 42. Kaiser, M.J.; Snyder, B.F. Modeling offshore wind installation costs on the U.S. *Outer Cont. Shelf. Renew. Energy* 2013, 50, 676–691. [CrossRef]
- 43. Topham, E.; McMillan, D. Sustainable decommissioning of an offshore wind farm. Renew. Energy 2017, 102, 470–480. [CrossRef]
- 44. Windeurope. Ports Support the Growth of the Wind Industry; WINDEUROPE: Brussels, Belgium, 2020.

- 45. González, M.M.; Trujillo, L. Reforms and infrastructure efficiency in Spain's container ports. *Transp. Res. Part A Policy Pract.* 2008, 42, 243–257. [CrossRef]
- 46. Marsh, G. UK ports: Ready for Round 3? Renew. Energy Focus 2012, 13, 38-41. [CrossRef]
- 47. Pavlic, B.; Cepak, F.; Sucic, B.; Peckaj, M.; Kandus, B. Sustainable port infrastructure, practical implementation of the green port concept. *Therm. Sci.* 2014, *18*, 935–948. [CrossRef]
- 48. Porter, A.; Phillips, S.U.S. West Coast Port Infrastructure Needs for Development of Floating Offshore Wind Facilities. In *Ports*; American Society of Civil Engineers: Reston, VA, USA, 2019; Volume 2019, pp. 366–375.
- 49. Rippel, D.; Jathe, N.; L'tjen, M.; Freitag, M. Evaluation of loading bay restrictions for the installation of offshore wind farms using a combination of mixed-integer linear programming and model predictive control. *Appl. Sci.* **2019**, *9*, 5030. [CrossRef]
- Scheltes, M. An Offshore Port Concept to Reduce the Construction Costs in Offshore Wind. Master's Thesis, Delft University of Technology, Delft, The Netherlands, 2018.
- Walther, L.; Schwientek, A. Logistic process analysis to develop the supply chain for offshore wind farm installations. In *Pioneering* Solutions in Supply Chain Performance Management: Concepts, Technology and Applications; Fraunhofer-Public: Munich, Germany, 2013; pp. 89–103.
- Ait-Alla, A.; Oelker, S.; Lewandowski, M.; Freitag, M.; Thoben, K.D. A study of new installation concepts of offshore wind farms by means of simulation model. In Proceedings of the 27th International Ocean and Polar Engineering Conference, San Francisco, CA, USA, 25–30 June 2017; pp. 607–612.
- 53. Backe, S. Strategic Optimization of Offshore Wind Farm Installation. Master's Thesis, University of Bergen, Bergen, Norway, 2017.
- 54. Beinke, T.; Ait Alla, A.; Freitag, M. Resource Sharing in the Logistics of the Offshore Wind Farm Installation Process based on a Simulation Study. *Int. J. e-Navig. Marit. Econ.* **2017**, *7*, 42–54. [CrossRef]
- 55. Bergqvist, R.; Monios, J. Green Ports in Theory and Practice. In *Green Ports Inland and Seaside Sustainable Transportation Strategies*; Elsevier: Amsterdam, The Netherlands, 2018; pp. 1–17. [CrossRef]
- Carbó, Y.D. Las Infraestructuras Para el Desarrollo de la Energía Eólica Marina en Cataluña. Master's Thesis, Escola Técnica Superior d'Enginyeria de Camins, Canals e Ports, UPC Barcelonatech, Barcelona, Spain, 2017.
- 57. Crown Estate Scotland. Ports for Offshore Wind; Crown Estate Scotland: Glasgow, Scotland, 2020.
- 58. D'Amico, F.; Mogre, R.; Clarke, S.; Lindgreen, A.; Hingley, M. How purchasing and supply management practices affect key success factors: The case of the offshore-wind supply chain. *J. Bus. Ind. Mark.* **2017**, *32*, 218–226. [CrossRef]
- Akbari, N. LEANWIND—Ports Suitability Assessment for Offshore Wind Development—Case Studies Report; WINDEUROPE: Brussels, Belgium, 2015.
- 60. Akbari, N.; Attari, A.; Cradden, L. Buoyant Gravity Based Foundations for Offshore Wind Infrastructure Challenges; EWEA: Brussels, Belgium, 2015.
- 61. Akbari, N. LEANWIND—Driving Cost Reductions in Offshore Wind; WINDEUROPE: Brussels, Belgium, 2017.
- 62. Castro, J.V.F.S. *Características de um Porto Para Suporte aos Empreendimentos Eólicos offshore: Proposta de Diretrizes;* Universidade Federal do Rio Grande do Norte: Natal, Brazil, 2019.
- 63. Castro-Santos, L.; Prado, G.; Costa, P.; Estanqueiro, A. Methodology to design an economic and strategic offshore wind energy Roadmap in Portugal. *Energy Procedia* **2013**, 1–11.
- 64. COWI North America Inc. Assessment of Ports and Infrastructure—NYSERDA; COWI North America Inc.: Lyngby, Denmark, 2017.
- 65. European Wind Energy Association. *Wind in our Sails the Coming of Europe's Offshore Wind Energy Industry;* European Wind Energy Association: Brussels, Belgium, 2011.
- 66. Garrad Hassan America Inc. Assessment of Ports for Offshore Wind Development in the United States; Garrad Hassan America Inc.: Austin, TX, USA, 2014.
- 67. Guillen, P.; Wetzler, N.; Abstoss, N. Analysis of Maryland Port Facilities for Offshore Wind Energy Services; Kinetik Partners, LLC.: Baltimore, MD, USA, 2011.
- Irawan, C.A.; Akbari, N.; Jones, D.F.; Menachof, D. A combined supply chain optimisation model for the installation phase of offshore wind projects. *Int. J. Prod. Res.* 2018, 56, 1189–1207. [CrossRef]
- 69. Irish Maritime Development Office. *IPORES 2018 A Review of Irish Ports Offshore Renewable Energy Services*; Irish Maritime Development Office: Dublin, Ireland, 2018.
- Jensen, M.B.; Kofoed-Hansen, H.; Hansen, A.; Christensen, L.M.; Jørgensen, M.H.; Hsu, M.A. Offshore Wind Port Feasibility Study of Taichung Harbor, Taiwan. 2016. Available online: https://www.academia.edu/33881798/Offshore_Wind_Port_Feasibility_ Study_of_Taichung_Harbor_Taiwan_Taichung (accessed on 5 January 2021).
- 71. Marsh, G. Vessel supply chain shapes up for offshore wind. Renew. Energy Focus 2010, 11, 18–23. [CrossRef]
- 72. Nilson, S.; Derugeris, J.; Hines, E.M.; Borkland, J.; Carchedi, D.; Baxter, D.Y. Port Infrastructure: Marine Commerce Terminal in New Bedford, MA. In *Ports* 2016; American Society of Civil Engineers: Reston, VA, USA, 2016; pp. 214–223. [CrossRef]
- Porter, A.; Phillips, S. Determining the Infrastructure Needs to Support Offshore Floating Wind and Marine Hydrokinetic Facilities on the Pacific West Coast and Hawaii. 2016. Available online: https://espis.boem.gov/final%20reports/5503.pdf (accessed on 18 May 2020).
- Snyder, B.; Kaiser, M.J. Ecological and economic cost-benefit analysis of offshore wind energy. *Renew. Energy* 2009, 34, 1567–1578. [CrossRef]
- 75. Tetra Tech EC Inc. Port and Infrastructure Analysis for Offshore Wind Energy Development; Tetra Tech EC Inc.: Bartow, FL, USA, 2010.

- 76. Ursavas, E. A benders decomposition approach for solving the offshore wind farm installation planning at the North Sea. *Eur. J. Oper. Res.* **2017**, *258*, 703–714. [CrossRef]
- 77. Wells, N.; McConnell, M. Assessment of the Irish Ports & Shipping Requirements for the Marine Renewable Energy Industry; Sustainable Energy Authority of Ireland and Irish Maritime Development Office: Dublin, Ireland, 2011.
- Whitney, P.R.; Wilson, S.J.K.; Chaston, S.; Elkinton, C.; Uriate, A. The Identification of Port Modifications and the Environmental and Socioeconomic Consequences the Identification of Port Modifications and the Environmental and Socioeconomic Consequences. OCS Study BOEM 2016.
- 79. BVG Associates. UK Ports for the Offshore Wind Industry: Time to Act; BVG Associates: Swindon, UK, 2009.
- 80. Byrne, D. Ship design trends and the implications for port design. In *Proceedings of the Institution of Civil Engineers-Civil Engineering;* Thomas Telford Ltd.: London, UK, 2010; Volume 163, pp. 41–48. [CrossRef]
- 81. Castro-Santos, L.; Filgueira-Vizoso, A.; Lamas-Galdo, I.; Carral-Couce, L. Methodology to calculate the installation costs of offshore wind farms located in deep waters. *J. Clean. Prod.* 2018, 170, 1124–1135. [CrossRef]
- 82. Esteban, M.D.; Couñago, B.; López-Gutiérrez, J.S.; Negro, V.; Vellisco, F. Gravity based support structures for offshore wind turbine generators: Review of the installation process. *Ocean Eng.* **2015**, *110*, 281–291. [CrossRef]
- Poulsen, T.; Chen, G.; Rytter, N.G.M. Wind energy shipping and logistics. In Proceedings of the International Conference on Logistics and Maritime Systems, Singapore, 12–14 September 2013; pp. 1–12.
- 84. Quandt, M.; Beinke, T.; Ait-Alla, A.; Freitag, M. Simulation Based Investigation of the Impact of Information Sharing on the Offshore Wind Farm Installation Process. *J. Renew. Energy* **2017**, 2017, 8301316. [CrossRef]
- 85. Castro-Santos, L.; Diaz-Casas, V. Life-cycle cost analysis of floating offshore wind farms. Renew. Energy 2014, 66, 41-48. [CrossRef]
- 86. Castro-Santos, L.; Garcia, G.P.; Simões, T.; Estanqueiro, A. Planning of the installation of offshore renewable energies: A GIS approach of the Portuguese roadmap. *Renew. Energy* **2019**, *132*, 1251–1262. [CrossRef]
- 87. Dalgic, Y.; Lazakis, I.; Dinwoodie, I.; McMillan, D.; Revie, M. Advanced logistics planning for offshore wind farm operation and maintenance activities. *Ocean Eng.* 2015, 101, 211–226. [CrossRef]
- Lütjen, M.; Karimi, H.R. Approach of a port inventory control system for the offshore installation of wind turbines. In Proceedings
 of the Twenty-second International Offshore and Polar Engineering Conference, Rhodes, Greece, 17–22 June 2012; pp. 502–508.
- 89. Beinke, T.; Freitag, M.; Zint, H.-P. Ressourcen-Sharing für eine bezahlbare Energiewende. *Betracht. Prod. Errichtungslogist.* Offshore-Windenerg. Ind. 40 Manag. 2015, 31, 7–11.
- Lange, K.; Rinne, A.; Haasis, H.D. Planning maritime logistics concepts for offshore wind farms: A newly developed decision support system. In Proceedings of the Computational Logistics: Third International Conference, ICCL 2012, Shanghai, China, 24–26 September 2012; pp. 142–158. [CrossRef]
- 91. Barlow, E.; Öztürk, D.T.; Revie, M.; Akartunalı, K.; Day, A.H.; Boulougouris, E. A mixed-method optimisation and simulation framework for supporting logistical decisions during offshore wind farm installations. *Eur. J. Oper. Res.* **2018**, *264*, 894–906. [CrossRef]
- 92. Faiz, T.I. Minimization of Transportation, Installation and Maintenance Operations Costs for Offshore Wind Turbines. Master's Thesis, Louisiana State University, Baton Rouge, LA, USA, 2014.
- 93. Grainger, B.; Thorogood, T. Beyond the harbor wall. IEE Rev. 2001, 2001, 13–17. [CrossRef]
- 94. Irawan, C.A.; Jones, D.; Ouelhadj, D. Bi-objective optimisation model for installation scheduling in offshore wind farms. *Comput. Oper. Res.* **2017**, *78*, 393–407. [CrossRef]
- 95. Kaiser, M.J.; Snyder, B. Modeling the decommissioning cost of offshore wind development on the U.S. Outer Continental Shelf. *Mar. Policy* **2012**, *36*, 153–164. [CrossRef]
- 96. Tekle Muhabie, Y.; Rigo, P.; Cepeda, M.; de Almeida D'Agosto, M.; Caprace, J.D. A discrete-event simulation approach to evaluate the effect of stochastic parameters on offshore wind farms assembly strategies. *Ocean Eng.* **2018**, *149*, 279–290. [CrossRef]
- Sarker, B.R.; Faiz, T.I. Minimizing transportation and installation costs for turbines in offshore wind farms. *Renew. Energy* 2017, 101, 667–679. [CrossRef]
- Yagi, K.; Tarutani, T.; Dohata, S.; Shiobara, Y. The First Offshore Wind Turbines in Setana Port in Japan. In Proceedings of the Oceans '04 MTS/IEEE Techno-Ocean, Kobe, Japan, 9–12 November 2004; pp. 1841–1846.
- 99. Kim, W.S.; Kim, J. Simulation models for offshore port service concepts. Appl. Sci. 2019, 9, 584. [CrossRef]
- 100. Irawan, C.A.; Wall, G.; Jones, D. An optimisation model for scheduling the decommissioning of an offshore wind farm. *OR Spectr.* **2019**, *41*, 513–548. [CrossRef]
- Calderon, R.A.C. Feasibility of a Marina Port Along the Buenos Aires Coast, Argentina. Master's Thesis, Delft University of Technology, Delft, The Netherlands, 2012.
- Hattam, C.; Hooper, T.; Papathanasopoulou, E. A well-being framework for impact evaluation: The case of the UK offshore wind industry. *Mar. Policy* 2017, 78, 122–131. [CrossRef]
- 103. Wurzel, R.K.W.; Moulton, J.F.G.; Osthorst, W.; Mederake, L.; Deutz, P.; Jonas, A.E.G. Climate pioneership and leadership in structurally disadvantaged maritime port cities. *Environ. Politics* **2019**, *28*, 146–166. [CrossRef]
- Ait-Alla, A.; Quandt, M.; Lütjen, M. Aggregate Installation Planning of Offshore Wind Farms. *Recent Adv. Electr. Comput. Eng.* 2013, 7, 23–30.
- 105. Andersen, P.D.; Clausen, N.E.; Cronin, T.; Piirainen, K.A. The North Sea Offshore Wind Service Industry: Status, perspectives and a joint action plan. *Renew. Sustain. Energy Rev.* 2018, *81*, 2672–2683. [CrossRef]

- 106. Barlow, E.; Ozturk, D.T.; Revie, M.; Akartunali, K.; Day, A.; Boulougouris, E. On Using Simulation to Model the Installation Process Logistics for an Offshore Wind Farm; unpublished; University of Strathclyde, Glasgow: Glasgow, Scotland, 2017.
- 107. Beinke, T.; Quandt, M.; Ait-Alla, A.; Freitag, M. The impact of information sharing on installation processes of offshore wind farms—Process modelling and simulation-based analysis. *Int. J. Shipp. Transp. Logist.* **2020**, *12*, 117–145. [CrossRef]
- Bellsolà Olba, X.; Daamen, W.; Vellinga, T.; Hoogendoorn, S.P. State-of-the-art of port simulation models for risk and capacity assessment based on the vessel navigational behaviour through the nautical infrastructure. *J. Traffic Transp. Eng.* 2018, *5*, 335–347. [CrossRef]
- 109. Castro-Santos, L.; Diaz-Casas, V.; Brage, R.Y. The importance of the activity costs in a shipyard: A case study for floating offshore wind platforms. *Ships Offshore Struct.* **2020**, *15*, 53–60. [CrossRef]
- 110. Castro-Santos, L.; Martins, E.; Guedes Soares, C. Methodology to calculate the costs of a floating offshore renewable energy farm. *Energies* **2016**, *9*, 324. [CrossRef]
- 111. Duin, H.; Gorldt, C.; Thoben, K.D.; Pawar, K. Process innovation through digital services for the seaport of the future. In Proceedings of the 2016 International Conference on Engineering, Technology and Innovation/IEEE International Technology Management Conference (ICE/ITMC), Trondheim, Norway, 13–15 June 2016; IEEE: New York, NY, USA, 2016; pp. 1–8. [CrossRef]
- McAuliffe, F.D.; Lynch, K.; Sperstad, I.B.; Nonas, L.M.; Halvorsen-Weare, E.E.; Jones, D.; Akbar, N.; Wall, G.; Irawan, C.; Norstad, I. The LEANWIND suite of logistics optimisation and full lifecycle simulation models for offshore wind farms. *J. Phys. Conf. Ser.* 2018, 1104, 012002. [CrossRef]
- Oelker, S.; Lewandowski, M.; Ait Alla, A.; Ohlendorf, J.-H.; Haselsteiner, A.F. Logistikszenarien für die Errichtung von Offshore-Windparks. Herausford. Wirtsch. Neuer. Logistikkonzepte Ind. 40 Manag. 2017, 33, 24–28.
- 114. Poulsen, T.; Hasager, C.B.; Jensen, C.M. The role of logistics in practical levelized cost of energy reduction implementation and government sponsored cost reduction studies: Day and night in offshorewind operations and maintenance logistics. *Energies* **2017**, *10*, 464. [CrossRef]
- Scholz-Reiter, B.; Lütjen, M.; Heger, J.; Schweizer, A. Planning and control of logistics for offshore wind farms. *Int. Conf. Math. Comput. Methods Sci. Eng. Proc.* 2010, 242–247. Available online: https://www.researchgate.net/publication/259889712 (accessed on 3 September 2020).
- 116. Scholz-Reiter, B.; Heger, J.; Lütjen, M.; Schweizer, A. A milp for installation scheduling of offshore wind farms. *Int. J. Math. Models Methods Appl. Sci.* 2011, *5*, 371–378.
- 117. Ahsan, D.; Pedersen, S. The influence of stakeholder groups in operation and maintenance services of offshore wind farms: Lesson from Denmark. *Renew. Energy* **2018**, *125*, 819–828. [CrossRef]
- 118. Shafiee, M.; Brennan, F.; Espinosa, I.A. A parametric whole life cost model for offshore wind farms. *Int. J. Life Cycle Assess.* 2016, 21, 961–975. [CrossRef]
- 119. Strid, J. Future Offshore Wind Farm Construction and Maintenance Strategies. Master's Thesis, Lund University, Lund, Sweden, 2017.
- 120. Bradley, D.; Peterson, J.; Koujak, D.D. Regional collaboration opportunities as catalyst to unleash U.S. offshore wind industry. *Electr. J.* **2014**, *27*, 28–34. [CrossRef]
- 121. Chartron, S. Improving logistics scheduling and operations to support offshore wind construction phase. *Logist. Res.* 2019, 12, 1–14. [CrossRef]
- 122. Ioannou, A.; Angus, A.; Brennan, F. A lifecycle techno-economic model of offshore wind energy for different entry and exit instances. *Appl. Energy* 2018, 221, 406–424. [CrossRef]
- 123. Ioannou, A.; Angus, A.; Brennan, F. Parametric CAPEX, OPEX, and LCOE expressions for offshore wind farms based on global deployment parameters. *Energy Sources Part B Econ. Plan Policy* **2018**, *13*, 281–290. [CrossRef]
- 124. Shih, Y. A note of wind engineering: Describing tower pre-assembly work at quayside for offshore wind turbine. *Energy Rep.* **2020**, *6*, 577–588. [CrossRef]
- Snyder, B.; Kaiser, M.J. A comparison of offshore wind power development in europe and the U.S.: Patterns and drivers of development. *Appl. Energy* 2009, *86*, 1845–1856. [CrossRef]
- 126. Beinke, T.; Ait Alla, A.; Freitag, M. Resource and information sharing for the installation process of the offshore wind energy. *IFIP Adv. Inf. Commun. Technol.* **2017**, *514*, 268–275. [CrossRef]
- 127. Lacal-Arántegui, R.; Yusta, J.M.; Domínguez-Navarro, J.A. Offshore wind installation: Analysing the evidence behind improvements in installation time. *Renew. Sustain. Energy Rev.* 2018, 92, 133–145. [CrossRef]
- 128. Leithead, W.; Camciuc, A.; Amiri, A.K.; Carroll, J. The X-Rotor Offshore Wind Turbine Concept. J. Phys. Conf. Ser. 2019, 1356, 012031. [CrossRef]
- 129. Aksoy, S.; Durmusoglu, Y. Improving competitiveness level of Turkish intermodal ports in the Frame of Green Port Concept: A case study. *Marit. Policy Manag.* 2019, 47, 203–220. [CrossRef]
- Gutierrez-Romero, J.E.; Esteve-Pérez, J.; Zamora, B. Implementing Onshore Power Supply from renewable energy sources for requirements of ships at berth. *Appl. Energy* 2019, 255, 113883. [CrossRef]
- Akbari, N.; Jones, D.; Treloar, R. A cross-European efficiency assessment of offshore wind farms: A DEA approach. *Renew. Energy* 2020, 151, 1186–1195. [CrossRef]
- 132. Halvorsen-Weare, E.E.; Gundegjerde, C.; Halvorsen, I.B.; Hvattum, L.M.; Nonås, L.M. Vessel fleet analysis for maintenance operations at offshore wind farms. *Energy Procedia* 2013, *35*, 167–176. [CrossRef]

- 133. Stefanakou, A.A.; Nikitakos, N.; Lilas, T.; Pavlogeorgatos, G. A GIS-based decision support model for offshore floating wind turbine installation. *Int. J. Sustain. Energy* **2019**, *38*, 673–691. [CrossRef]
- 134. Cabral, T.P. Porto-Indústria e Clusters Portuários: Uma Análise Comparativa. Bachelor's Thesis, Universidade Federal de Santa Catarina, Florianópolis, Brazil, 2014.
- 135. Grant, D.B.; Williams, T.; Bishop, D.; Menachof, D. Capabilities of Logistics and Ports in the Humber Region. In Proceedings of the 19th Annual Logistics Research Network, Huddersfield, UK, 3–5 September 2014; p. 8.
- Gupta, A.K.; Gupta, S.K.; Patil, R.S. Environmental management plan for port and harbour projects. *Clean Technol. Environ. Policy* 2005, 7, 133–141. [CrossRef]
- 137. Rippel, D.; Jathe, N.; Lütjen, M.; Freitag, M. BAse-port capacity optimization for the installation of offshore wind farms. *Simul. Ser.* **2020**, *52*, 32–43.
- 138. Regt, D.N. Economic Feasibility of Offshore Service Locations for Maintenance of Offshore Wind Farms on the Durch Part of the North Sea. Ph.D. Thesis, Delft University of Technology, Delft, The Netherland, 2012.
- Backe, S.; Haugland, D. Strategic Optimization of Offshore Wind Farm Installation. In Proceedings of the Computational Logistics: 8th International Conference, ICCL 2017, Southampton, UK, 18–20 October 2017; pp. 285–299. [CrossRef]
- Rippel, D.; Jathe, N.; Lütjen, M.; Freitag, M. A mixed-integer formulation to optimize the resupply of components for the installation of offshore wind farms. In Proceedings of the 2020 Winter Simulation Conference (WSC), Orlando, FL, USA, 14–18 December 2020; pp. 1420–1431.
- 141. Stanković, J.J.; Marjanović, I.; Papathanasiou, J.; Drezgić, S. Social, economic and environmental sustainability of port regions: Mcdm approach in composite index creation. *J. Mar. Sci. Eng.* **2021**, *9*, 74. [CrossRef]
- 142. Raileanu, A.B.; Onea, F.; Rusu, E. Implementation of offshore wind turbines to reduce air pollution in coastal areas-case study constanta harbour in the black sea. *J. Mar. Sci. Eng.* **2020**, *8*, 550. [CrossRef]
- 143. Acciaro, M.; Ghiara, H.; Cusano, M.I. Energy management in seaports: A new role for port authorities. *Energy Policy* 2014, 71, 4–12. [CrossRef]
- 144. Satir, T.; Doğan-Sağlamtimur, N. The protection of marine aquatic life: Green Port (EcoPort) model inspired by Green Port concept in selected ports from Turkey, Europe and the USA. *Period. Eng. Nat. Sci.* **2018**, *6*, 120–129. [CrossRef]
- 145. de Moura, D.A.; de Andrade, D.G. Concepts of green port operations—One kind of self-diagnosis method to the port of Santos—Brazil. *Indep. J. Manag. Prod.* 2018, *9*, 785–809. [CrossRef]
- Asgari, N.; Hassani, A.; Jones, D.; Nguye, H.H. Sustainability ranking of the UK major ports: Methodology and case study. *Transp. Res. Part E Logist. Transp. Rev.* 2015, 78, 19–39. [CrossRef]
- Badurina, P.; Cukrov, M.; Dundović, Č. Contribution to the implementation of "Green Port" concept in Croatian seaports. Sci. J. Marit. Res. 2017, 31, 10–17. [CrossRef]
- 148. Oniszczuk-Jastrząbek, A.; Pawłowska, B.; Czermański, E. Polish sea ports and the Green Port concept. *SHS Web Conf.* **2018**, 57, 01023. [CrossRef]
- 4C Offshore. Hull Quays & Terminals. Available online: https://www.4coffshore.com/ports/port-infrastructure-hull-unitedkingdom-pid22.html (accessed on 3 February 2020).
- 150. 4C Offshore. MHI Vestas Schedules Testing. Available online: https://www.4coffshore.com/news/mhi-vestas-schedules-testing--nid1947.html (accessed on 3 February 2020).
- 151. 4C Offshore. Wind Lifts Port of Esbjerg's 2017. Available online: https://www.4coffshore.com/news/wind-lifts-port-of-esbjerg2 7s-2017-nid7289.html (accessed on 3 February 2020).
- 152. ABP. Location Hull. Available online: https://www.abports.co.uk/locations/hull/ (accessed on 3 February 2020).
- 153. BIS. Offshore Terminal Bremerhaven; BIS: Bremerhaven, Germany, 2011.
- 154. BIS. Wind City Map: Offshore Starts in Bremerhaven; BIS: Bremerhaven, Germany, 2013.
- 155. BIS. Offshore Terminal Bremerhaven; BIS: Bremerhaven, Germany, 2017; pp. 1–15.
- 156. BIS Bremerhaven. Ports and Logistics. Available online: https://www.bis-bremerhaven.de/business-location/ports-and-logistics.98402.html (accessed on 3 September 2020).
- 157. Bremenports. Offshore-Terminal Bremerhaven. Available online: https://www.bremenports.de/en/ports/bremerhaven (accessed on 3 September 2020).
- 158. BIS Bremerhaven Offshore Wind Port. Overview of bremerhaven's ports and their infrastructure. Available online: https://www.bis-bremerhaven.de/en/an-ocean-of-benefits-for-business/ports-and-logistics (accessed on 3 September 2020).
- 159. Green Port Hull. Projects. Available online: https://greenporthull.co.uk/projects (accessed on 4 September 2020).
- Green Port Hull. Learning and Training. Available online: https://greenporthull.co.uk/jobs-training/learning-and-training (accessed on 5 September 2020).
- 161. Green Port Hull. About Green Port. Available online: https://greenporthull.co.uk/about-green-port (accessed on 5 September 2020).
- Green Port Hull. Job Roles in Offshore Wind. Available online: https://greenporthull.co.uk/jobs-training/job-roles-in-offshorewind-1 (accessed on 5 September 2020).
- 163. Greenport. A Commitment to Biodiversity. Available online: https://www.greenport.com/news101/europe/a-commitment-tobiodiversity (accessed on 5 September 2020).

- 165. Greenport. Green Hydrogen Study for German Ports. Available online: https://www.greenport.com/news101/lng/greenhydrogen-study-for-german-ports (accessed on 28 November 2020).
- 166. Greenports. Sustainability Report 2017/2018. Available online: https://sms.bremenports.de/storm2microsite/report/sustainability-report-20172018 (accessed on 28 November 2020).
- 167. Greenports. Umweltbericht 2020 für die Häfen Bremen/Bremerhaven 171; Greenports: Bremerhaven, Germany, 2020.
- 168. Port of Esbjerg. Accelerating the Green Transition. Available online: https://portesbjerg.dk/en/news/nu-gar-det-staerkt-i-dengronne-omstilling (accessed on 23 December 2020).
- 169. Port of Esbjerg. Port Facilities; Esbjerg Havn: Esbjerg, Denmark, 2017.
- 170. Port of Esbjerg. Annual Report Port of Esbjerg; Esbjerg Havn: Esbjerg, Denmark, 2018.
- 171. Port of Esbjerg. Port Esbjerg—Sustainability Strategy—Environment, Climate and Occupational Health and Safety; Esbjerg Havn: Esbjerg, Denmark, 2019.
- 172. Port of Esbjerg. Port Esbjerg Offshore Wind Hub; Esbjerg Havn: Esbjerg, Denmark, 2020.
- 173. Port of Esbjerg. Port of Esbjerg—Offshore Wind Port and Logistic; Esbjerg Havn: Esbjerg, Denmark, 2020.
- 174. Port of Esbjerg. Port Details & Services; Esbjerg Havn: Esbjerg, Denmark, 2020.
- 175. SEARATES. Port of Bremerhaven (Germany). Available online: https://www.searates.com/port/bremerhaven_de.htm (accessed on 27 December 2020).
- 176. SEARATES. Port of Esbjerg (Denmark). Available online: https://www.searates.com/port/esbjerg_dk.htm (accessed on 27 December 2020).
- 177. SEARATES. Port of Hull (United Kingdom). Available online: https://www.searates.com/port/hull_gb.htm (accessed on 27 December 2020).
- 178. THE POWER HUB. Ready to Take Offshore—Europe's Premier Address for the Offshore Wind Energy Industry. 2012. Available online: https://www.bremerhaven.de/sixcms/media.php/94/Final_Broschuere-Stark-am-Strom_EN.pdf (accessed on 27 December 2020).
- 179. González, M.; Santiso, A.; Jones, D.; Akbari, N.; Vasconcelos, R.; Melo, D. Offshore Wind Power Growth and Industrial Development in Emerging Markets. *Energies* **2024**, *17*, 4712. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.